Author's response to reviews

**Title:** The effectiveness of a suicide prevention app for Indigenous Australian youths: a randomised controlled trial

**Authors:**

Fiona L Shand (fionas@unsw.edu.au)
Rebecca Ridani (r.ridani@unsw.edu.au)
Joseph Tighe (joe.akg@bigpond.com)
Helen Christensen (h.christensen@unsw.edu.au)

**Version:** 2  **Date:** 16 September 2013

**Author's response to reviews:** see over
16 September 2013

The Editor
Trials Journal

Dear editor

Re: MS 1759968620976848

Thank you kindly for considering the above manuscript. We have now had the opportunity to revise the manuscript in response to the reviewer’s comments, attached.

We look forward to your decision.

Yours sincerely

Fiona Shand (on behalf of all authors)
Response to reviewer 1

Minor Essential Revisions

_The manuscript will benefit from thorough editing_

Thank you to the reviewer – we have thoroughly edited the manuscript.

_Level of interest: An article of importance in its field_

We thank the reviewer for this comment.

_Needs some language corrections before being published_

We have edited the manuscript and hope the language is now satisfactory.

_Statistical review: No, the manuscript does not need to be seen by a statistician._